Author's response to reviews

Title: Increases in summer serum 25-hydroxyvitamin D (25OHD) concentrations in elderly subjects in Sao Paulo, Brazil vary with age, gender and ethnicity

Authors:

Sergio S Maeda (ssetsuo@terra.com.br)
Ilda S Kunii (ilda.kunii@unifesp.br)
Lilian F Hayashi (lilian.hayashi@unifesp.br)
Marise Lazaretti-Castro (lazaretti.castro@unifesp.br)

Version: 4 Date: 3 May 2010

Author's response to reviews: see over
Dear reviewer
Denise von Muhlen:

We would like to thank you again for your interest and comments which certainly improved our manuscript. Below you will find our response to each item:

1) Page 2- Abstract
   a) Suggestion accepted and modified in the manuscript
   b) We accepted your explanation
2) Corrected
3) Corrected
4) Suggestion accepted and modified in the manuscript
5) Suggestion accepted and modified in the manuscript
6) Suggestion accepted and modified in the manuscript
7) Suggestion accepted and modified in the manuscript

São Paulo, May 3rd, 2010

Sincerely,

Sergio Setsuo Maeda, M.D.
Marise Lazaretti Castro, MD, Ph.D.